Author's response to reviews

Title: The relationship between distal radius fracture malunion and arm-related disability: A prospective population-based cohort study with 1-year follow-up

Authors:

Elisabeth Brogren (elisabeth.brogren@skane.se)
Manfred Hofer (manfred.hofer@skane.se)
Michael Petranek (michael.petranek@skane.se)
Philippe Wagner (pw@nko.se)
Lars B Dahlin (lars.dahlin@med.lu.se)
Isam Atroshi (isam.atroshi@skane.se)

Version: 2 Date: 27 May 2010

Author's response to reviews:

Dear Sir,

I mistakenly inserted the study’s Ethical Committee approval number as Trial Registration Number. The study is a prospective cohort study and was not registered in any Trial Registries. This has been changed in the abstract.

Acknowledgements have been added.

I hope everything is right this time.

Sincerely yours,

Elisabeth Brogren